

G OPEN ACCESS

Citation: Camara M, Ouattara E, Duvignaud A, Migliani R, Camara O, Leno M, et al. (2017) Impact of the Ebola outbreak on *Trypanosoma brucei gambiense* infection medical activities in coastal Guinea, 2014-2015: A retrospective analysis from the Guinean national Human African Trypanosomiasis control program. PLoS Negl Trop Dis 11(11): e0006060. https://doi.org/10.1371/ journal.pntd.0006060

Editor: Philippe Büscher, Institute of Tropical Medicine, BELGIUM

Received: February 1, 2017

Accepted: October 20, 2017

Published: November 13, 2017

Copyright: © 2017 Camara et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The database used for this work is available from the Dryad Digital Repository datadryad.org https://doi.org/10.5061/ dryad.036ds.

Funding: Data collection was part of the Guinean national HAT control program activities held by the Guinean government with the financial contribution of the Foundation for Innovative New Diagnostics

RESEARCH ARTICLE

Impact of the Ebola outbreak on *Trypanosoma brucei gambiense* infection medical activities in coastal Guinea, 2014-2015: A retrospective analysis from the Guinean national Human African Trypanosomiasis control program

Mariame Camara^{1®}, Eric Ouattara^{2,3,4®}, Alexandre Duvignaud^{2,3,4®}*, René Migliani², Oumou Camara¹, Mamadou Leno¹, Philippe Solano⁵, Bruno Bucheton^{5‡}, Mamadou Camara^{1‡}, Denis Malvy^{2,3,4‡}

1 Programme National de Lutte contre la Trypanosomiase Humaine Africaine PNLTHA-Ministère de la Santé, Conakry, République de Guinée, 2 Univ. Bordeaux, Inserm, Infectious Diseases in Resource Limited Countries, U1219, ISPED, Bordeaux, France, 3 Department of Tropical Medicine and Clinical International Health, CHU Bordeaux, Bordeaux, France, 4 Programme PAC-CI/ANRS Research Site, CHU Treichville, Abidjan, Côte d'Ivoire, 5 UMR 177 IRD-CIRAD INTERTRYP, IRD, Montpellier, France

• These authors contributed equally to this work.

‡ BB, MC and DM also contributed equally to this work.

* alex.duvignaud@gmail.com

Abstract

Background

The 2014–2015 Ebola outbreak massively hit Guinea. The coastal districts of Boffa, Dubreka and Forecariah, three major foci of Human African Trypanosomiasis (HAT), were particularly affected. We aimed to assess the impact of this epidemic on sleeping sickness screening and caring activities.

Methodology/Principal findings

We used preexisting data from the Guinean sleeping sickness control program, collected between 2012 and 2015. We described monthly: the number of persons (i) screened actively; (ii) or passively; (iii) treated for HAT; (iv) attending post-treatment follow-up visits. We compared clinical data, treatment characteristics and Disability Adjusted Life-Years (DALYs) before (February 2012 to December 2013) and during (January 2014 to October 2015) the Ebola outbreak period according to available data. Whereas 32,221 persons were actively screened from February 2012 to December 2013, before the official declaration of the first Ebola case in Guinea, no active screening campaigns could be performed during the Ebola outbreak. Following the reinforcement and extension of HAT passive surveillance system early in 2014, the number of persons tested passively by month increased from 7 to 286 between April and September 2014 and then abruptly decreased to 180 until January 2015 and to none after March 2015. 213 patients initiated HAT treatment, 154 (72%) before Ebola and 59 (28%) during the Ebola outbreak. Those initiating HAT therapy during Ebola



(passive surveillance), the Bill and Melinda Gates Foundation (vector control), and the World Health Organization (HAT treatment). This study received no dedicated funding. MC received a travel grant from the French government to attend a medical training course at Bordeaux University Hospital. AD received a travel grant from the European Society of Clinical Microbiology and Infectious Diseases (https://www.escmid.org/) to present preliminary results of this work (abstract no. 6921) as a poster during the 27th ECCMID, 22-25th April 2017, Vienna, Austria. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

outbreak were recruited through passive screening and diagnosed at a later stage 2 of the disease (96% vs. 55% before Ebola, p<0.0001). The proportion of patients attending the 3 months and 6 months post-treatment follow-up visits decreased from 44% to 10% (p<0.0001) and from 16% to 3% (p = 0.017) respectively. The DALYs generated before the Ebola outbreak were estimated to 48.7 (46.7–51.5) and increased up to 168.7 (162.7–174.7), 284.9 (277.1–292.8) and 466.3 (455.7–477.0) during Ebola assuming case fatality rates of 2%, 5% and 10% respectively among under-reported HAT cases.

Conclusions/Significance

The 2014–2015 Ebola outbreak deeply impacted HAT screening activities in Guinea. Active screening campaigns were stopped. Passive screening dramatically decreased during the Ebola period, but trends could not be compared with pre-Ebola period (data not available). Few patients were diagnosed with more advanced HAT during the Ebola period and retention rates in follow-up were lowered. The drop in newly diagnosed HAT cases during Ebola epidemic is unlikely due to a fall in HAT incidence. Even if we were unable to demonstrate it directly, it is much more probably the consequence of hampered screening activities and of the fear of the population on subsequent confirmation and linkage to care. Reinforced program monitoring, alternative control strategies and sustainable financial and human resources allocation are mandatory during post Ebola period to reduce HAT burden in Guinea.

Author summary

This work was conducted in coastal Guinea, the last focus in Western Africa where the transmission of Human African Trypanosomiasis (HAT) is still very active. The Guinean government and his partners are conducting HAT control activities to reduce the burden of this neglected tropical disease and, as set-up by WHO, to eliminate it as a public health problem by 2020. Unfortunately, control efforts were deeply impaired during the Ebola outbreak that stroke the country in 2014–15. The aim of the study was to evaluate the impact of this unprecedented outbreak on HAT screening and caring activities. A major impact was the interruption of all active screening activities which aim, in addition of detecting and treating patients, is to clear the human reservoir of parasite and decrease transmission. Passive surveillance and diagnosis were also severely affected due to the fear of the population to come to the endemic district confirmation and treatment facilities and to the progressive banning of using rapid test in peripheral health structures, as well as the requisition of healthcare workers and facilities for the fight against Ebola. As a consequence, only 59 HAT patients were diagnosed and treated during the Ebola outbreak (from January 2014 to October 2015) as compared to 154 before the outbreak (from February 2012 to December 2013). This potentially high number of undiagnosed human reservoir of trypanosomes may have contributed in turn to increase transmission levels. A rapid revival of HAT control activities in Guinea is thus vital in order to stay in line with the 2020 elimination goal and to avoid possible bursts of the disease.

Introduction

Human African trypanosomiasis (HAT) is a vector born parasitic disease due to the infection by *T. brucei gambiense* in Western and Central Africa and *T. brucei rhodesiense* in Eastern Africa. In its late stage, HAT is generally considered to be fatal if left untreated and is responsible for a heavy burden in affected areas. It is one of the most neglected tropical diseases and, until recently, the treatment relied on ancient and toxic drugs [1]. In Western Africa, most cases of *T. brucei gambiense* infection are found in remote sparse active focuses such as the three major ones situated in the mangrove ecosystem of coastal Guinea[2], namely from North to South: Boffa [3–4], Dubreka [4] and Forecariah [4–6]. Of note, these three districts were among the most affected by the epidemic of Ebola virus disease that severely hit Guinea in 2014–2015 [7].

Previous studies have demonstrated the impact of the recent Ebola outbreak on health systems of affected countries with an unprecedented magnitude on HIV care or malaria control activities in Guinea and Liberia [8–13]. First, the Ebola crisis has weakened an already fragile healthcare system by the re-allocation of the available resources to fight Ebola. In addition, health-care workers who were in first-line have paid a heavy toll with 1.5% deaths among doctors, nurses, and midwives compared to 0.02% in the general population of Guinea [14]. Second, Ebola burden has led to a drastic reduction in healthcare center attendances [9]. The high transmission rates of Ebola virus in healthcare centers had raised a justifiable fear against the health system in general with dramatic consequences [15,16].

Since HAT control activities are generally vulnerable and poorly supported in terms of human or financial resources, one might expect that the Ebola epidemic has deeply impaired HAT control as well. Therefore, we aimed at investigating the impact of the Ebola outbreak on HAT screening and caring activities in the three coastal Guinean focuses of Boffa, Dubreka and Forecariah.

Methods

Settings

The Guinean national HAT control program is devoted to reducing the burden of sleeping sickness through a comprehensive action: (i) fight against the vector with targeted vector control using insecticide impregnated targets; (ii) improving human infection diagnosis by reinforcing passive screening with the implementation of new rapid diagnostic tests at peripheral level (health posts and centers) and conducing targeted active screening campaigns; (iii) treatment and follow-up of patients diagnosed with HAT. Our analysis focuses on screening and caring (HAT diagnosis confirmation and subsequent treatment and follow-up) purposes.

Our study was carried out in three districts of coastal Guinea, namely from north to south Boffa, Dubreka and Forecariah (Fig 1). The Boffa [3,4] and Dubreka [4] districts are located at the north of the capital city Conakry and Forecariah [4–6] district is located at the south of Conakry. All districts are situated in a mangrove ecosystem favoring the growth of the HAT vector population. The area mainly hosts Soussou and Baga ethnic groups. Inhabitants have outside activities such as irrigated rice growth, salt extraction, wood cut, and fishing [3] exposing them to high contacts with tsetse flies. In 2005, the prevalence of HAT in the area was estimated to be about 1% [17]. The number of newly reported HAT cases was 78 in 2013 and increased to 106 cases in 2016[18].

An unprecedented Ebola outbreak struck Guinea from December 2013 to April 2016 with more than 3,800 persons infected of whom 2,500 died [19]. The epidemic started in South East Guinea region and reached Conakry and coastal Guinea around March 2014. Many healthcare centers including HAT treatment centers were requisitioned for the fight against Ebola (Fig 2).

PLOS Neglected Tropical Diseases | https://doi.org/10.1371/journal.pntd.0006060 November 13, 2017

4/15

Sierra Leone 20,0 - 29.9National road 30.0 - 99.9Head town of prefecture 100.0 et plus Head town of region 50 100.Km Fig 1. Ebola virus disease and HAT treatment centers spatial distribution. This figure shows the spatial distribution of Ebola virus disease (EVD) incidence with both EVD and Human African Trypanosomiasis (HAT) treatment centers and newly diagnosed HAT cases during the study period (before

and during Ebola outbreak) in coastal Guinea. *HAT: Human African Trypanosomiasis. ¥ Before Ebola: from Feb.2012 to Dec.2013. § During Ebola: from Jan.2014 to Oct.2015.

https://doi.org/10.1371/journal.pntd.0006060.g001

Study and population

We performed a retrospective analysis of the data of all the people screened and/or treated for HAT in the three districts of Boffa, Dubreka and Forecariah within the frame of the Guinean HAT National Control Program (PNLTHA) routine activities. For that purpose, we used available data collected between February 2012 and October 2015 in two electronic databases: (i)



NEGLECTED TROPICAL DISEASES PLOS



Fig 2. Timeline of the Ebola epidemic and HAT control activities. This figure shows in parallel the main events of the Ebola epidemic and of Human African Trypanosomiasis (HAT) control activities between October 2013 and April 2016.

https://doi.org/10.1371/journal.pntd.0006060.g002

the 'HAT treatment database' collecting prospectively sociodemographic, clinical, parasitological, therapeutic and follow-up information of the patients initiating HAT treatment in the study centers between January 2012 and October 2015; (ii) the 'HAT screening database' recording active testing campaigns activities since 2012 and passive screening activities only since 2014. Regarding Ebola, we used previously published data reporting monthly cases in Guinea between March 2014 and October 2015 [19]. Data deposited in the Dryad repository, https://doi.org/10.5061/dryad.036ds [20]. HAT-infected persons were recruited during active screening campaigns throughout the entire districts or by passive detection in persons presenting spontaneously to the three HAT treatment centers based in the Forecariah, Dubreka and Boffa cities and, as from 2014, some people presenting at peripheral health centers of the same districts who underwent decentralized screening before the interdiction of blood sampling due to Ebola. Indeed, early in 2014, HAT passive detection capacities were reinforced by the implementation of a passive surveillance system in which SD-Bioline rapid diagnostic tests [21] were made available at the level of the endemic district's peripheral health centers (approximately 30 per disease focus). Additional cases were also actively searched by targeted screening of the neighborhood of previous cases. Positive serological suspects were referred to the endemic district laboratories for parasitological study and if confirmed, treatment was performed. Individuals with positive Card Agglutination Test for Trypanosomiasis (CATT) underwent lymph node puncture when relevant and /or blood sampling for direct diagnostic confirmation according to previously described procedures [22]. Cerebrospinal fluid collection by lumbar puncture in confirmed cases allowed disease staging and treatment guidance [22]. Patients with 5 or less white blood cells (WBC)/ml of CSF were considered as stage 1 HAT patients whereas those with 6 or more cells were classified as stage 2 patients. The persons diagnosed with HAT were admitted in HAT treatment centers situated close to the

endemic district laboratories. Boffa, Dubreka and Forecariah centers have a capacity of respectively 17 beds, 16 beds and 4 beds. After a treatment preparation phase consisting in pain relief, deworming and administration of an anti-malaria treatment, pentamidine isethionate (4 mg/ kg) was administered intramuscularly during 7 days to patients at the stage 1 of the disease. The NECT regimen (nifurtimox at the dosage of 15 mg/kg/day divided in three oral intakes for 10 days and eflornithine 400 mg/kg/day administered in two intravenous infusions for 7 days) was used for patients with stage 2 disease. Post-treatment follow-up visits were planned 3, 6 and 12 months after treatment completion according to national procedures applicable at that time.

Outcomes and statistical analysis

We evaluated the impact of Ebola outbreak on HAT screening activities by analyzing overtime the number of persons screened during active campaign and the number of persons screened passively during the 2014–2015 period (data were not available for passive screening before 2014).

To evaluate the impact of Ebola outbreak on caring activities, we first analyzed overtime the overall number of patients initiating HAT treatment and the number patients who attended the follow-up visits 3 and 6 months after HAT treatment was completed. Then, we compared sociodemographic, clinical, treatment and follow-up variables before and during Ebola outbreak using Fisher exact tests. The period referred as 'before Ebola period' ran from February 2012 (beginning of available data in the "HAT treatment database") to December 2013 (the first cases of Ebola appeared in late december 2013 in "Guinée forestière") and the 'Ebola period' extended from January 2014 (first plain month after the first Ebola cases in Guinea and beginning of the "HAT screening database") to October 2015 (end of available data).

We estimated Disability Adjusted Life-Years (DALYs) before and after the Ebola crisis, using the same methodology than the 2010 Global Burden of Disease study, without application of age-weighting and discounting [23]. We weighted average DALYs for the following 3 sub-groups: (i) reported HAT cases before Ebola; (ii) reported cases during Ebola; (iii) underreported cases during Ebola. The latter group refers to HAT cases that were not reported during the Ebola period compared to what has been observed before the Ebola outbreak, assuming that HAT incidence remained stable during Ebola and the decrease in the number of newly diagnosed HAT cases was attributable mainly to Ebola outbreak. Those assumptions are in line with previous reports from WHO [18]. Reported deaths and life expectancy at the age of death were used to estimate the Years of Life Lost (YLLs) in both sub-groups (i) and (ii). Under-reported deaths were generated from binomial distributions with assumptive case fatality rates of 2%, 5% and 10% corresponding to conservative, average and pessimistic scenarios. Years Lived with Disability (YLDs) were estimated using disability weighting of 0.21 for early disease stage, 0.35 for late advanced stage; treatment duration of 1 month and time to death for untreated patients at 3 years [23]. We computed 95% confidence intervals around estimates using bootstrap (1,000 patients resampling, for detailed methodology see S1 Text).

Statistical analyses were performed using SAS 9.3 (SAS Institute, Cary, North Carolina, USA).

Ethics

All HAT patients described in the framework of this study were diagnosed and treated according to the national health and WHO policy. Agreement was obtained from the Ministry of Health of Guinea to use the National Control Program database (2012–2016) to assess the impact of Ebola on sleeping sickness. Data were anonymized prior to the analysis. All the investigations were conducted in accordance with the declaration of Helsinki. This work fulfills the STROBE criteria.

Results

Active screening campaigns (2012–2015)

Fig 3A shows the monthly number of persons screened during active campaigns. Overall 32,221 persons were screened for HAT during 6 active campaigns led since February 2012. The last active screening survey was performed in December 2013 during the follow-up of previously treated patients at their home. A total of 291 family members or neighbors were tested for HAT during this activity. No other active screening campaign was led out after that, during the Ebola period.

Among the persons screened in the 2012–2013 period, 26,950 (84%), where from the Boffa district as the result of an elimination project initiated in 2012 in this area[24]. Three lesser campaigns were led in September 2013 with 2,810 persons screened in Dubreka, in October 2013 and November 2013 with respectively 2,299 and 453 persons screened in Forecariah.

Passive screening activities (2014–2015)

Overall 1,836 persons were tested passively for HAT between January 2014 and October 2015. Fig 3B shows the monthly number of persons tested during this period. Passive testing at peripheral heath centers was performed at a low level in Dubreka and Forecariah districts from



Fig 3. Impact of Ebola outbreak on HAT testing and caring activities in Guinea, January 2012 to October 2015. This panel figure displays: (A) Monthly evolution of the number of persons diagnosed during active campaigns by HAT treatment center; (B) Monthly evolution of the number of persons tested passively by HAT treatment center (and as from 2014 corresponding district); (C) Monthly evolution of the number of persons initiating therapy by HAT treatment center; (D) Monthly evolution of the number of persons initiating HAT therapy according to the type of screening; (E) Monthly evolution of the number of persons initiating HAT therapy by disease stage at diagnosis (F) Monthly evolution of the number of persons attending 3 months post-treatment follow-up visit. HAT: Human African Trypanosomiasis: *Passive routine testing data were available only between January 2014 and October 2015. **All post-treatment follow-up visits were centralized at the Dubreka center whatever the place where the patients received HAT therapy (Dubreka, Boffa or Forecariah HAT centers).

https://doi.org/10.1371/journal.pntd.0006060.g003

January 2014 to April 2014 as the passive surveillance system was reinforced. In Boffa, activities started slightly later in May 2014. After April 2014, the number of persons tested passively rapidly increased to reach a peak of 286 persons in September 2014. After September 2014, the number of tests abruptly decreased to a plateau of about 180 persons until January 2015. From there, the number of tests rapidly decreased to no testing after February 2015 in Forecariah and after March 2015 in Boffa and Dubreka with the interdiction to perform Rapid Detection Test-ing (RDT) in peripheral health centers. The referral rate of individuals testing positive at a health center was also very low as only 18% (13 of 70 serological suspects) were indeed seen for confirmation at the HAT treatment centers (of whom 7 were confirmed as HAT patients). Unfortunately, we were not able to compare this parameter with pre-Ebola period.

Treatment and follow-up (2012–2015)

Between February 2012 and October 2015, 213 patients were treated for HAT, 154 (72%) before and 59 (28%) during the Ebola outbreak in coastal Guinea (Fig 1). Of note, all patients with confirmed HAT initiated the treatment. During the whole study period, the mean number of HAT patients treated monthly was 5 (Confidence Interval 95% (CI95%): 3–7). This number was 7 (95%CI: 3–11) before the Ebola period and 3 (95%CI: 2–3) during the Ebola period (S1 Table). The number of patients treated for HAT was the highest in March 2012 and March 2013, with 35 and 30 patients respectively, corresponding to large active screening campaigns led in the Boffa focus (Fig 3C, Table 1).

Among these patients, 104 (49%) were recruited during active campaigns and 107 (51%) through passive screening (Fig 3D, Table 1). All the patients initiating the treatment during Ebola period were recruited passively (100% vs 32% before Ebola, p<0.0001) (Fig 3D, Table 1).

Regarding the activities of the different treatment centers, 143 (67%), 48 (23%), and 22 (10%) treatments were initiated at Dubreka, Boffa and Forecariah, respectively. The proportion of patient treated in the Dubreka center significantly increased during the Ebola period (75% vs 64% before Ebola, p = 0.048) (Fig 1, Fig 3C, Table 1) as the Forecariah and Boffa centers were requisitioned in May and September 2014 respectively. Overall, 130 patients (61%) were male and 161 (76%) were >18 years old. Of 201 patients with available data, 115 (57%) had inside occupational activities in mangrove area. During EVD period, significantly less patients aged < 18 years old initiated treatment (14% vs 28% before EVD, p = 0.031) (Table 1).

During the Ebola period, significantly more patients were diagnosed with stage 2 disease compared to the pre-Ebola period (96% vs. 81%, p = 0.0039, Table 1). If we consider missing values (n = 26) as failure to complete HAT therapy, more patients did not complete their treatment during Ebola than before (25% vs. 11%, p = 0.0129). Of note, the 2 (1%) patients who died during the study period were admitted in a comatose state and passed before NECT could be initiated. The proportion of patients attending the 3 months and 6 months post-treatment follow-up visits (centralized at the Dubreka center) decreased from 44% to 10% (p<0.0001) and from 16% to 3% (p = 0.017) respectively, before and during Ebola period.

To account for uncertainty concerning the split date chosen (January 2014) between pre-Ebola and Ebola periods, we performed a sensitivity analysis with two different split dates (December 2013 and March 2014). The results remained similar (see <u>S2 Table</u> and <u>S3 Table</u>).

Disability Adjusted Life Years (DALYs) estimates

Only 1 death was reported for both periods, i.e.; 1 (1%) and 1 (2%) before and during Ebola respectively (Table 1). The 2 patients who died were admitted in a comatose presentation and passed out before the therapeutic regimen could be administered. Hence, assuming a case

Table 1. Characteristics of the patients treated for HAT before and during Ebola outbreak, Guinea, January 2012 to October 2015 (for sensitivity analysis with different split dates between the two periods see S2 Table and S3 Table).

NE	Both periods	Before Ebola outbreak (23 months)	During Ebola outbreak (22 months)	P value
13	213 (100%)	154 (72%)	59 (28%)	-
12				0.530
	130 (61%)	96 (63%)	34 (58%)	
	82 (39%)	57 (37%)	25 (42%)	
12				0.031
	51 (24%)	43 (28%)	8 (14%)	
	161 (76%)	110 (72%)	51 (86%)	
13				0.048
	48 (23%)	41 (27%)	7 (12%)	
	143 (67%)	99 (64%)	44 (75%)	
	22 (10%)	14 (9%)	8 (13%)	
01				0.874
	73 (36%)	54 (37%)	19 (35%)	
	13 (7%)	9 (6%)	4 (7%)	
	115 (57%)	83 (57%)	32 (58%)	
11				<0.0001
	107 (51%)	49 (32%)	58 (100%)	
	104 (49%)	104 (68%)	0 (0%)	
13				<0.0039
	31 (15%)	29 (19%)	2 (4%)	
	182 (85%)	125 (81%)	57 (96%)	
13				0.0129
	179 (96%)	136 (88%)	43 (73%)	
	32 (15%)	17(11%)	15 (25%)	
	2 (1%)	1 (1%)	1 (2%)	
13				<0.0001
	139 (65%)	86 (56%)	53 (90%)	
	74 (35%)	68 (44%)	6 (10%)	
13		· · ·		0.017
	187 (88%)	130 (84%)	57 (97%)	
	26 (12%)	24 (16%)	2 (3%)	
	N 13 112 1 112 1 112 1 112 1 113 1 114 1 115 1 115	Both periods 13 213 (100%) 12 1 12 1 130 (61%) 82 (39%) 112 1 141 1 151 (24%) 1 161 (76%) 1 143 (67%) 1 143 (67%) 1 143 (67%) 1 13 1 13 (7%) 1 115 (57%) 1 114 (49%) 1 107 (51%) 1 113 (7%) 1 114 (107 (51%) 1 115 (57%) 1 114 (107 (51%) 1 115 (57%) 1 116 (107 (51%) 1 117 (96%) 1 13 (15%) 1 143 (179 (96%) 1 13 (179 (96%) 1 13 (139 (65%) 1 139 (65%) 1 139 (65%) 1 139 (65%) 1 137 (88%) <	N Both periods Before Ebola outbreak (23 months) 13 213 (100%) 154 (72%) 112	Both periods Before Ebola outbreak (22 months) During Ebola outbreak (22 months) 13 213 (100%) 154 (72%) 59 (28%) 12

HAT: Human African Trypanosomiasis

⁽¹⁾ HAT treatment centers were health centers specialized in the care of HAT

⁽²⁾ Outside rural activity (farmers, woodcutters, and fishermen); Outside city activity include (taxi drivers, mechanics, house builders, black-smith, and carpenters), inside activity (student, office employees, plumbers, tailors, and photographs)

⁽³⁾ Patients with 5 or less white blood cells (WBC)/ml of CSF were considered as stage 1 HAT patients whereas those with 6 or more white blood cells (WBC)/ml of CSF were classified as stage 2 HAT patients

⁽⁴⁾ Treatment not completed include 4 patients (3 before, 1 during Ebola) transferred to other centers, 2 patients (both before Ebola) who left the healthcare centers without medical authorization and 26 patients with missing data.

https://doi.org/10.1371/journal.pntd.0006060.t001

fatality rate (CFR) of 2% (similar to what has been observed among HAT cases reported during the Ebola period) among the under-reported cases (cases that should have been reported based on previous reports), the estimated number of deaths increased from 1 before Ebola to 3 during Ebola (Table 2). In alternative scenarios based on CFR of 5% and 10%, the number of

	Before Ebola outbreak (n = 154 reported cases)	During Ebola outbreak (n = 59 reported and 95 under-reported cases) ⁽¹⁾			
		Conservative scenario (Under- reported CFR = 2%)	Average scenario (Under- reported CFR = 5%)	Pessimistic scenario (Under- reported CFR = 10%)	
Number of deaths (2)	1.0	3.0	6.0	10.0	
Years Lost due to Disability (YLD) ⁽³⁾	4.0	101.4	101.4	101.4	
Years of Life Lost (YLL) (4)	44.7 (42.0–47.5)	116.1 (110.8–121.4)	232.2 (225.0–239.7)	413.7 (403.5–423.9)	
Disability-Adjusted Life Year (DALY) ⁽⁴⁾	48.7 (46.7–51.5)	217.5 (212.2–222.7)	333.6 (326.3–340.9)	515.1 (504.9–525.2)	
Overall DALY increase (During Ebola—Before Ebola)	-	168.7 (162.7–174.7)	284.9 (277.1–292.8)	466.3 (455.7–477.0)	

Table 2. Disability-Adjusted Life Years (DALY) estimates for HAT cases before and during Ebola outbreak, Guinea (January 2012 to October 2015).

CFR: Case Fatality Rate

⁽¹⁾ The number of HAT cases reported was 154 and 59 before and during Ebola respectively. The number of reported cases over the 23 months before Ebola period was consistent with the previous WHO reports [18]. We then assumed that 95 HAT cases, the difference between the cases reported before and during Ebola were under-reported during Ebola period.

⁽²⁾ The number of deaths in under-reported HAT cases was generated from a binomial distribution. The presented results are rounded.

⁽³⁾ The 95% confidence intervals are not shown because of very small variations

⁽⁴⁾ Values are mean values (with lower and upper 95% confidence intervals) from 1,000 bootstrap simulations

https://doi.org/10.1371/journal.pntd.0006060.t002

deaths during Ebola increased to 6 and 10 respectively. YLLs was 44.7 (42.0–47.5) before Ebola and increased after Ebola up to 116.1 (110.8–121.4), 232.2 (225.0–239.7) and 413.7 (403.5–423.9) for under-reported CFR of 2%, 5% and 10% respectively. The YLD estimate increased from 4.0 before Ebola to 101.4 during Ebola for all scenarios. Finally, the DALYs were estimated to 48.7 (46.7–51.5) before Ebola and increased to 168.7 (162.7–174.7), 284.9 (277.1–292.8) and 466.3 (455.7–477.0) during Ebola assuming a CFR in under-reported HAT cases of 2%, 5% and 10% respectively (Table 2, for detailed estimations see S4 Table).

Discussion

Our study suggests a major impact of the Ebola outbreak on HAT screening activities in the Western Africa most active focuses of Boffa, Dubreka and Forecariah in costal Guinea. All active screening campaigns had to be abandoned during the Ebola outbreak and passive screening activities were also impacted despite new efforts to avail rapid diagnostic tests in peripheral health centers in active foci. The number of person screened passively at health centers was first reduced by half during the early phase of the Ebola outbreak to zero after the outbreak peak. Nevertheless, some patients kept coming by themselves to the only remaining functional treatment center in Dubreka throughout the Ebola period. Even if we were not able to demonstrate a direct impact of Ebola on the referral rate of serological suspects (people with a positive screening test) to HAT centers for diagnostic confirmation and subsequent treatment, the number of patients initiating HAT therapy was reduced by 2/3 during Ebola, with 96% of them diagnosed at a late stage of the disease and a poor post-treatment retention in follow-up. It is unlikely that this drop could be explained only by a fall in HAT incidence. Hence, Ebola outbreak contributed to worsening the burden of HAT in Guinea, given DALYs were increasing from 48.7 to 217.5 in our conservative scenario. This increase in DALYs might have

reached up to 466 in the most pessimistic scenario assuming a CFR of 10% among underreported HAT cases during Ebola outbreak, in line with previous data from WHO [25].

Active screening campaigns, aimed at reducing the human reservoir of trypanosomes are very important for the control of HAT in the hardly reachable mangrove ecosystem of costal Guinea where access to healthcare is limited. The mass screening campaign planned in the Boffa focus in March 2014 had to be abandoned during its awareness phase after the first Ebola cases have been officially declared in Guinea. Prior to the Ebola period, such campaigns provided 68% of the patients treated for HAT and enabled the detection/treatment of persons in the early disease stage 1, when no or only none-specific symptoms are present. Targeting these individuals is of up-most importance as, remaining active, they represent a major source of infection for the vector [26], and because the treatment outcome is by far much favorable. Initially, for evident safety reasons and to avoid population gatherings, and then because of population mistrust toward the health system, all active screening activities had to be abandoned throughout the Ebola period, thereby hindering the effort to reduce the T. brucei gambiense reservoir and global HAT burden through early diagnosis and treatment. Only 2 (4%) patients were diagnosed in stage 1 during the Ebola period as compared to 29 (19%) before Ebola (Table 1). This sharp deficit highlights the fact that an important number of tsetse infective individuals were left undiagnosed during the Ebola outbreak. It is yet too early to assess precisely the impact of this on T. brucei gambiense transmission levels, but it is noteworthy that 79 HAT patients were recently diagnosed in the one and only Boffa focus where active screening activities could recently be implemented again (May and October 2016). This sharp increase of HAT prevalence, reaching up to 5% in several villages of the focus, is probably largely attributable to the three-year absence of active screening in these areas (data from Guinean national HAT control plan, to be published later).

With the inability to perform active screening, all medical activities to control HAT in costal Guinea during the Ebola period had to rely on the passive detection of patients. Although the data on the number of persons passively tested for HAT in the three endemic focuses were scarcely available before 2014, passive cases (self-presenting at HAT treatment centers) represented 32% of HAT patients in the pre-Ebola period (Table 1). Data on passive screening were computerized and systematically collected after 2014 when the capacities of Boffa, Dubreka and Forecariah health districts were reinforced by the settlement of a passive surveillance system in which serological suspicion was made at the level of local health centers (# 30 per disease focus) and parasitological confirmation was done at the district treatment centers. The scale-up of reinforced passive screening activities has unfortunately coincided with the Ebola outbreak resulting in a reduction by half in the number of persons tested during the early epidemic. This can be partly explained by the fear of the population to attend health structures, leading to a sharp attendance decrease. In this regard, it is noteworthy that less than 20% (13/ 70) of the persons who tested positive to the HAT rapid test at the peripheral health center level indeed showed up at one of the districts confirmation laboratories. This phenomenon was amplified by the fact that the HAT confirmation/treatment centers of Forecariah and Boffa were requisitioned early during the Ebola outbreak leaving Dubreka as the only functional center. Hence the mean distance to seek for HAT confirmation and treatment was increased and discouraged many clinical suspects. The use of rapid tests in peripheral health centers was then progressively abandoned with the interdiction of taking capillary blood samples in these areas.

Interestingly, significantly less people aged < 18 years were diagnosed with HAT during Ebola (14% vs. 28%, p = 0.031). Maybe parents were less prone to let their children go outside (or to accompany them to health centers) during this period.

Although late stage diagnosis is a classical feature of HAT passive diagnosis [4], the proportion of patients diagnosed through passive screening with stage 2 disease has increased sharply from 57 to 97% during the Ebola period (S5 Table). This illustrates another indirect impact of Ebola on HAT which was to increase the time length between infection and diagnosis in passively screened patients. Even with the well tolerated NECT regimen, the treatment of advanced infection remains hazardous [27]: greater efforts are required to raise the patient's health status before initiating anti-trypanosome therapy; advanced stage patients require a constant attention from the medical teams and they often need to stay under observation for several days/weeks after treatment completion before they can be discharged. HAT diagnosis and treatment are provided freely in Guinea. Nevertheless, increase of hospitalization length can represent a heavy economic burden for affected families as food is not supplied and the patients need guardians to look after them. Economic losses are particularly high when the patient's homes are located far away from the treatment centers. This was the case for many patients during the Ebola period as Dubreka became quickly the only available HAT treatment center in Guinea. Another impact of the fear of Ebola among communities on the patient care is that the proportion of patients coming for follow-up visits after treatment was significantly reduced, even if the new WHO 2014 guidelines which were no longer recommending systematic follow-up visits for HAT treated patients had not yet been implemented in Guinea.

Assessing the impact of Ebola on HAT public health outcomes was not possible with the data available. However, it is noteworthy that no difference was observed in the lethality for the reported cases denoting the tremendous efforts of the medical staff to maintain acceptable treatment condition despite facing major challenges. DALYs estimates using the available data have shown a significant impact of Ebola. Even if these estimates rely on many assumptions due to the lack of evidence, the impact of Ebola outbreak was already substantial with the conservative hypothesis of a similar 2% CFR for reported and under-reported HAT cases during Ebola. Few studies have estimated DALYs for HAT infection, mainly addressing *T. brucei rhodesiense* sleeping sickness [28,29]. We however, used comparable DALYs estimation methods and were more conservative about the CFR in under-reported cases in accordance with the lower morbidity of *T. brucei gambiense* sleeping sickness [28–30].

Our study has several limitations. First, we used distinct databases with different periods of data collection for active screening, passive screening and treatment. Thus, we were not able to statistically compare the trends in screening activities variation between periods, especially for passive screening. The result of passive screening activities should be interpreted with caution since the improvement of data collection may have induced reporting bias. In addition, results on screening activities before Ebola indirectly derived from the data of the patients treated for HAT may be subject to selection bias. Second, we have no data on adverse events during HAT treatment or post treatment sequelae which combined with mortality may have better captured treatment outcome. Third, our analysis was limited to before and during Ebola periods since the data for post Ebola period were not fully available. Hence, we certainly not have captured all the burden due to HAT cases left undiagnosed, since *T. brucei gambiense* HAT is a medium-long term evolution disease. But the DALYs estimation is a way of appreciating this burden despite this limit (even if subject to approximations). Post-Ebola data should comfort our findings by showing expected increasing HAT transmission after Ebola epidemic (preliminary data from Guinean national plan against HAT, to be published later).

Our findings are consistent with previous studies on the impact of Ebola on HIV care in forest Guinea and Liberia [9,11] and indicate an important impact also on HAT control and patient care in costal Guinea. Importantly, it is very likely that the deficit of active screening campaigns and the progressive failure of HAT passive detection have left many undiagnosed *T. brucei gambiense* infected persons in endemic areas. This may have created in turn favorable

epidemiological conditions for disease burden enhancement in some areas. Increased awareness efforts toward HAT endemic communities as well as the revival of the passive surveillance system together with large active screening campaigns allowing early HAT diagnosis and treatment are thus crucial and timely to more fully evaluate the impact of Ebola on HAT transmission and avoid possible dramatic bursts of disease prevalence in endemic foci. Targeted vector control measures, as those implemented previously in part of the Boffa focus [24] may also help speeding–up the reduction of transmission levels and stay in line with the 2020 elimination goal [31].

Finally, to limit the consequences of such a crisis in the future and make HAT control programs more resilient, it appears crucial that "less vulnerable" strategies such as vector control be generalized and reinforced, alongside with an ambitious and sustainable (in terms of human and financial resources allocation) active screening campaigns program, in order to reduce both transmission and human reservoir. As for Ebola, the involvement of community leaders could probably be a good way of increasing both awareness and adhesion of the population, as well as limiting the costs, thereby improving the efficiency of the entire program.

Supporting information

S1 Text. Disability Adjusted Life-Years (DALY) calculation. (DOCX)

S1 Table. Mean number of HAT patients treated before and during Ebola outbreak, Guinea (January 2012 to October 2015). (DOCX)

S2 Table. Characteristics of patients treated for HAT before and during Ebola outbreak, Guinea (January 2012 to October 2015), *Cutoff December 2013*. (DOCX)

S3 Table. Characteristics of patients treated for HAT before and during Ebola outbreak, Guinea (June 2012 to October 2015), <u>Cutoff March 2014</u>. (DOCX)

S4 Table. Disability-Adjusted Life Years (DALY) estimates before and during Ebola outbreak, Conakry Guinea 2012–2015 (Extended results). (DOCX)

S5 Table. HAT disease clinic stage before and during Ebola outbreak in patients detected through passive testing, Guinea (2012 to 2015). (DOCX)

Acknowledgments

The authors want to acknowledge all the people taking part into the Guinean national program against HAT activities or who are working in the dedicated treatment centers.

Author Contributions

Conceptualization: Mariame Camara, Eric Ouattara, Alexandre Duvignaud, Bruno Bucheton, Mamadou Camara, Denis Malvy.

Data curation: Mariame Camara, René Migliani, Oumou Camara, Mamadou Leno, Bruno Bucheton, Mamadou Camara.

- Formal analysis: Eric Ouattara, Alexandre Duvignaud, René Migliani, Bruno Bucheton, Denis Malvy.
- Methodology: Mariame Camara, Eric Ouattara, Alexandre Duvignaud, René Migliani, Bruno Bucheton, Denis Malvy.
- **Project administration:** Mariame Camara, Oumou Camara, Mamadou Leno, Philippe Solano, Bruno Bucheton, Mamadou Camara.
- Supervision: Philippe Solano, Bruno Bucheton, Mamadou Camara, Denis Malvy.
- Writing original draft: Mariame Camara, Eric Ouattara, Alexandre Duvignaud, René Migliani, Bruno Bucheton, Mamadou Camara, Denis Malvy.
- Writing review & editing: Mariame Camara, Eric Ouattara, Alexandre Duvignaud, René Migliani, Bruno Bucheton, Mamadou Camara, Denis Malvy.

References

- Brun R, Blum J, Chappuis F, Burri C. Human African trypanosomiasis. Lancet 2010; 375: 148–159. https://doi.org/10.1016/S0140-6736(09)60829-1 PMID: 19833383
- 2. WHO | Report of the second WHO stakeholders meeting on gambiense human African trypanosomiasis elimination. In: WHO [Internet]. [cited 17 Oct 2017]. Available: http://www.who.int/trypanosomiasis_african/resources/9789241511520/en/
- Kagbadouno MS, Camara M, Rouamba J, Rayaisse J-B, Traoré IS, Camara O, et al. Epidemiology of sleeping sickness in Boffa (Guinea): where are the trypanosomes? PLoS Negl Trop Dis. 2012; 6: e1949. https://doi.org/10.1371/journal.pntd.0001949 PMID: 23272259
- Vanhecke C, Guevart E, Ezzedine K, Receveur M-C, Jamonneau V, Bucheton B, et al. [Human African trypanosomiasis in mangrove epidemiologic area. Presentation, diagnosis and treatment in Guinea, 2005–2007]. Pathol Biol (Paris). 2010; 58: 110–116. <u>https://doi.org/10.1016/j.patbio.2009.07.033</u> PMID: 19854583
- Ilboudo H, Jamonneau V, Camara M, Camara O, Dama E, Léno M, et al. Diversity of response to Trypanosoma brucei gambiense infections in the Forecariah mangrove focus (Guinea): perspectives for a better control of sleeping sickness. Microbes Infect Inst Pasteur. 2011; 13: 943–952. <u>https://doi.org/10. 1016/j.micinf.2011.05.007</u> PMID: 21658462
- Courtin F, Jamonneau V, Camara M, Camara O, Coulibaly B, Diarra A, et al. A geographical approach to identify sleeping sickness risk factors in a mangrove ecosystem. Trop Med Int Health. 2010; 15: 881– 889. https://doi.org/10.1111/j.1365-3156.2010.02559.x PMID: 20545924
- 7. Republic of Guinea: Ministry of Health, World Health Organization. Guinea—EBOV [Internet]. [cited 20 Jan 2016]. Available: http://guinea-ebov.github.io/
- Moisan F, Traore A, Zoumanigui D, Feindouno JY, Sagno AM, Mollet T, et al. Public health structures attendance during the Ebola outbreak in Guéckédou, Guinea. Epidemiol Infect. 2016; 144: 2338–2344. https://doi.org/10.1017/S0950268816000728 PMID: 27086773
- Leuenberger D, Hebelamou J, Strahm S, De Rekeneire N, Balestre E, Wandeler G, et al. Impact of the Ebola epidemic on general and HIV care in Macenta, Forest Guinea, 2014. AIDS 2015; 29: 1883–1887. https://doi.org/10.1097/QAD.00000000000784 PMID: 26372393
- Parpia AS, Ndeffo-Mbah ML, Wenzel NS, Galvani AP. Effects of Response to 2014–2015 Ebola Outbreak on Deaths from Malaria, HIV/AIDS, and Tuberculosis, West Africa. Emerg Infect Dis. 2016; 22: 433–441. https://doi.org/10.3201/eid2203.150977 PMID: 26886846
- Loubet P, Mabileau G, Baysah M, Nuta C, Taylor M, Jusu H, et al. Likely effect of the 2014 Ebola epidemic on HIV care in Liberia. AIDS 2015; 29: 2347–2351. https://doi.org/10.1097/QAD. 00000000000821 PMID: 26544705
- Walker PGT, White MT, Griffin JT, Reynolds A, Ferguson NM, Ghani AC. Malaria morbidity and mortality in Ebola-affected countries caused by decreased health-care capacity, and the potential effect of mitigation strategies: a modelling analysis. Lancet Infect Dis. 2015; 15: 825–832. https://doi.org/10.1016/ S1473-3099(15)70124-6 PMID: 25921597
- Plucinski MM, Guilavogui T, Sidikiba S, Diakité N, Diakité S, Dioubaté M, et al. Effect of the Ebola-virusdisease epidemic on malaria case management in Guinea, 2014: a cross-sectional survey of health facilities. Lancet Infect Dis. 2015; 15: 1017–1023. https://doi.org/10.1016/S1473-3099(15)00061-4 PMID: 26116183

- Evans DK, Goldstein M, Popova A. Health-care worker mortality and the legacy of the Ebola epidemic. Lancet Glob Health. 2015; 3: e439–440. https://doi.org/10.1016/S2214-109X(15)00065-0 PMID: 26163833
- Kim HS, Sherman DK, Updegraff JA. Fear of Ebola: The Influence of Collectivism on Xenophobic Threat Responses. Psychol Sci. 2016; 27: 935–944. https://doi.org/10.1177/0956797616642596 PMID: 27207872
- Anoko JN. Communication during an outbreak of Ebola Virus Disease with rebellious communities in Guinea [Internet]. [cited 2 Nov 2017]. Available: http://www.ebola-anthropology.net/wp-content/ uploads/2014/12/Communicationduring-an-outbreak-of-Ebola-Virus-Disease-with-rebelliouscommunities-in-Guinea.pdf
- Camara M, Kaba D, KagbaDouno M, Sanon JR, Ouendeno FF, Solano P. [Human African trypanosomiasis in the mangrove forest in Guinea: epidemiological and clinical features in two adjacent outbreak areas]. Médecine Trop 2005; 65: 155–161.
- GHO | By category | Number of new reported cases (T.b. gambiense)—Data by country. In: WHO [Internet]. [cited 15 Jun 2017]. Available: http://apps.who.int/gho/data/node.main.A1636?lang=en
- Migliani R, Keïta S, Diallo B, Mesfin S, Perea W, Dahl B, et al. [Epidemiological aspects of Ebola virus disease in Guinea (december 2013-april 2016)]. Bull Soc Pathol Exot 1990. 2016; 109: 218–235. https://doi.org/10.1007/s13149-016-0511-4 PMID: 27456159
- 20. Camara M, Ouattara E, Duvignaud A, Migliani R, Camara O, Leno M, Solano P, Bucheton B, Camara M, Malvy D. Data from: Impact of the Ebola outbreak on Trypanosoma brucei gambiense infection diagnosis and treatment activities in coastal Guinea, 2014–2015: retrospective analysis from the Guinean national Human African Trypanosomiasis control program. [Internet]. Dryad Digital Repository. Available: https://doi.org/10.5061/dryad.036ds
- Sternberg JM, Gierliński M, Biéler S, Ferguson MAJ, Ndung'u JM. Evaluation of the diagnostic accuracy of prototype rapid tests for human African trypanosomiasis. PLoS Negl Trop Dis. 2014; 8: e3373. https://doi.org/10.1371/journal.pntd.0003373 PMID: 25521120
- 22. Malvy D, Chappuis F. Sleeping sickness. Clin Microbiol Infect 2011; 17: 986–995. https://doi.org/10. 1111/j.1469-0691.2011.03536.x PMID: 21722252
- 23. WHO | Metrics: Disability-Adjusted Life Year (DALY). In: WHO [Internet]. [cited 15 Jun 2017]. Available: http://www.who.int/healthinfo/global_burden_disease/metrics_daly/en/
- Courtin F, Camara M, Rayaisse J-B, Kagbadouno M, Dama E, Camara O, et al. Reducing Human-Tsetse Contact Significantly Enhances the Efficacy of Sleeping Sickness Active Screening Campaigns: A Promising Result in the Context of Elimination. PLoS Negl Trop Dis. 2015; 9: e0003727. <u>https://doi.org/10.1371/journal.pntd.0003727 PMID: 26267667</u>
- Organization WH, WHO Expert Committee on the Control and Surveillance of Human African Trypanosomiasis (2013: Geneva S. Control and surveillance of human African trypanosomiasis: report of a WHO expert committee [Internet]. World Health Organization; 2013. Available: http://www.who.int/iris/ handle/10665/95732
- Berthier D, Brenière SF, Bras-Gonçalves R, Lemesre J-L, Jamonneau V, Solano P, et al. Tolerance to Trypanosomatids: A Threat, or a Key for Disease Elimination? Trends Parasitol. 2016; 32: 157–168. https://doi.org/10.1016/j.pt.2015.11.001 PMID: 26643519
- Eperon G, Balasegaram M, Potet J, Mowbray C, Valverde O, Chappuis F. Treatment options for second-stage gambiense human African trypanosomiasis. Expert Rev Anti Infect Ther. 2014; 12: 1407– 1417. https://doi.org/10.1586/14787210.2014.959496 PMID: 25204360
- Fèvre EM, Odiit M, Coleman PG, Woolhouse ME, Welburn SC. Estimating the burden of rhodesiense sleeping sickness during an outbreak in Serere, eastern Uganda. BMC Public Health. 2008; 8: 96. https://doi.org/10.1186/1471-2458-8-96 PMID: 18366755
- Robays J, Raguenaud ME, Josenando T, Boelaert M. Eflornithine is a cost-effective alternative to melarsoprol for the treatment of second-stage human West African trypanosomiasis in Caxito, Angola. Trop Med Int Health. 2008; 13: 265–271. https://doi.org/10.1111/j.1365-3156.2007.01999.x PMID: 18304274
- Odiit M, Coleman PG, Liu W-C, McDermott JJ, Fèvre EM, Welburn SC, et al. Quantifying the level of under-detection of Trypanosoma brucei rhodesiense sleeping sickness cases. Trop Med Int Health. 2005; 10: 840–849. https://doi.org/10.1111/j.1365-3156.2005.01470.x PMID: 16135190
- WHO | WHO to roll out implementation strategy to eliminate sleeping sickness. In: WHO [Internet]. [cited 24 Jan 2017]. Available: http://www.who.int/neglected_diseases/HAT_roll_out_strategy_2013/ en/